Author's response to reviews

Title: Proximal myopathy in lactovegetarian Asian patients responding to Vitamin D and calcium supplement therapy - Report of two cases and literature review

Authors:

Hood Thabit (hoodthabit@physicians.ie)  
Maurice Barry (mbarry@ireland.com)  
Seamus Sreenan (diabetes.centre@hse.ie)  
Diarmuid Smith (diarmuidsmith@beaumont.ie)

Version: 5 Date: 23 January 2011

Author's response to reviews: see over
23rd January 2011

RE: 1760106211357347 - Proximal myopathy in lactovegetarian Asian patients responding to Vitamin D and calcium supplement therapy - Report of two cases and literature review

Dear Editor,

We would like to thank the reviewers of the above manuscript for their constructive comments and we have now modified the paper in response to the reviewer’s comment. We feel that the paper is significantly strengthened as a result and would like to resubmit it for considerations for publication in the Journal of Medical Case Reports.

The following reply has been made, based on the reviewer’s comment below:

**Reviewer’s Comment**

Introduction line 6: It should read "..vitamin D deficiency induced myopathy" not "...vitamin D induced myopathy".

The reviewer’s correction has been noted and this has been rectified in the revised manuscript.